Point-by-Point Letter response to Peer Review Remarks

Dear Editors and peer-reviewers,

Thank you for your remarks concerning the submitted manuscript under revision **ESPS manuscript NO:** 26605

Title: The Use of Rifaximin in Gastrointestinal and Liver Diseases

Reviewer's code: 00535896

Please find below the point-by-point changes:

- 1. The title has not been altered since we believe this is the most fitting for our manuscript
- 2. an Open Access excerpt and "© **The Author(s) 2015**. Published by Baishideng Publishing Group Inc. All rights reserved" excerpt has been added and highlighted in accordance to the manuscript guidelines
- 3. An Audio core tip meeting the requirements has been created and uploaded
- 4. The in text figure has been deleted and added to the last "Figure Legends" page
- 5. Based on Dr. Ya-Juan Ma's suggestions
 - a. Page 5 "diarrhea predominant IBD" has been corrected into "diarrhea predominant IBS"
 - b. Page 6 "SIBO" has been defined
 - c. Undefined letters on pages 14 and 18 have been correctly defined.

Thank you,
Ala Sharara, M.D., FACG, AGAF
Professor and Head, Division of Gastroenterology
American University of Beirut Medical Center
Beirut, Lebanon
Consulting Professor
Duke University Medical Center

P.O Box 11-0236/16B Beirut, Lebanon 11072020 +961-1-350000 extension 5345/5350